Author’s response to reviews

Title: Payment for performance (P4P): any future in Italy?

Authors:

Silvana Castaldi (silvana.castaldi@unimi.it)
Annalisa Bodina (annalisa.bodina@inwind.it)
Luciana Bevilacqua (luciana.bevilacqua@ospedaleniguarda.it)
Elena Parravicini (elena.parravicini@ospedaleniguarda.it)
Michaela Bertuzzi (michaela.bertuzzi@ospedaleniguarda.it)
Francesco Auxilia (francesco.auxilia@unimi.it)

Version: 5 Date: 15 April 2011

Author’s response to reviews: see over
Dear Mrs Audrey Ann Reyes and Dr. Natalie Pafitis MSc,
As you asked in the mail on 01.04.2011 the quality of written English has been improved (through a commercial copyediting service). We have highlighted all changes made. In the tables we have replaced commas with points.

Best Regards
Silvana Castaldi